\begin{table}%t4
\caption{\label{long}Long GRBs with measured redshifts and spectral parameters not
already in the sample of 83~bursts considered by~N08.  References: (1)~Golenetskii et~al. (\cite{Golenetskii2008}a); (2)~Ohno et~al. (\cite{Ohno2008}); (3) Barthelmy et~al. (\cite{Barthelmy2008}); (4) Tueller et~al. (\cite{Tueller2008}); (5) Golenetskii et~al. (\cite{Golenetskii2008}a), GCN~7854; (6) Golenetskii et~al. (\cite{Golenetskii2008}b); (7) Golenetskii et~al. (\cite{Golenetskii2008}c); (8) Meegan et~al. (\cite{Meegan2008}a); (9) Baumgartner et~al. (\cite{Baumgartner2008}).}
%\centerline
 {\small \begin{tabular}{lllllllllll}
\hline\hline
\noalign{\smallskip}
GRB  & $z$  &    $\alpha$   &  Peak flux & Range   &  $L_{\rm iso}$ &$E_{\rm peak}$&  Fluence & Range   &  $E_{\rm iso}$   &  Ref.   \\
 &          &               &            & (keV)   &  $10^{52}$~erg/s  &  (keV)  &  ($10^{-6}$) & keV     &  $10^{52}$~erg   & \\
\hline
080411 & 1.03 & --1.51(0.05) & 1.3(0.2)e-5& 20--2000 & 9.4(0.4) & 526(63) & 63(3.1) & 20--2000 & 24(2)     &  1   \\
080413   & 2.433   & --1.2(0.1)     & 0.8(0.2)   & 15--1000 & 0.6(0.1)          &   584(206)   & 4.8(1.0)     & 15--1000 & 8.5(1.0)         &  2   \\
080413B  & 1.1     & --1.26(0.27)   & 18.7(0.8) &  15--150  & 1.7(0.3)          &   154(33)    & 3.2(0.1)     & 15--150  & 2.0(0.4)         &  3   \\        
080603B  & 2.69    & --1.21(0.3)    & 1.5(0.4)e-6& 20--1000 & 12(0.5)           &   376(76)    & 2.4(0.1)     & 20--1000 & 11(1.6)          &  4   \\
080605   & 1.639   & --1.03(0.07)   & 1.6(0.3)e-5& 20--2000 & 32(1.3)           &   252(19)    & 30.2(1.2)    & 20--2000 & 25.3(3.6)        &  5   \\
080607   & 3.036   & --1.08(0.07)   & 2.7(0.5)e-5& 20--4000 & 217(10)           &   1691(170)  & 89(5)        & 20--4000 & 200(13)          &  6  \\ 
080721$^a$& 2.591   & --0.9(0.1)     & 2.0(0.3)e-5& 20--5000 & 102(15)           &   1742(226)  & 84(6)        & 20--5000 & 120(12)          &  7   \\
080810   & 3.35    & --0.91(0.12)   & 1.9(0.2)   & 50--300  & 9.3(0.9)          &   1488(348)  & 6.9(0.5)     & 50--300  & 39(3.7)          &  8   \\
080916A  & 0.689   & --1.17(0.21)   & 2.7(0.2)   & 15--150  & 0.08(0.02)        &   161(39)    & 4.0(0.1)     & 15--150  & 1(0.2)           &  9  \\ 
\hline
\end{tabular}}
\medskip
$^a$~Band spectrum with $\beta=-2.43\pm0.35$.  
\end{table}